Author’s response to reviews

Title: Novel mutations of PKD genes in the Czech population with autosomal dominant polycystic kidney disease

Authors:

Veronika Elisakova (velisakova@post.cz)
Jitka Stekrova (jstek@lf1.cuni.cz)
Lena Obeidova (lena.obeidova@gmail.com)
Jana Reiterova (jreiterova@seznam.cz)
Miroslav Merta (miroslav.merta@lf1.cuni.cz)
Vladimir Tesar (vladimir.tesar@lf1.cuni.cz)
Frantisek Losan (genetika@genetika-plzen.cz)
Milada Kohoutova (mkoho@lf1.cuni.cz)

Version: 3 Date: 28 February 2014

Author’s response to reviews:

Dear Sir/Madam,

Please find revised manuscript “Novel mutations of PKD genes in the Czech population with autosomal dominant polycystic kidney disease” as an attachment. We would like to thank to all reviewers for their helpful comments and suggestions. The two comments of reviewer Fiona Karet have been fulfilled. Details of the cohort have been transferred to the Results section and language editing has been provided.

Best regards,

Veronika Elisakova